Author’s response to reviews

Title: Usability evaluation with mental health professionals and young people to develop an Internet-based cognitive-behaviour therapy program for adolescents with anxiety disorders

Authors:

Lori M Wozney (lori.wozney@iwk.nshealth.ca)
Pamela Baxter (BaxterP@mcmaster.ca)
Amanda S Newtwon (an6@ualberta.ca)

Version: 5 Date: 7 December 2015

Author's response to reviews: see over
December 7, 2015

**Re: MS 1054571571142856**

*Usability evaluation with adolescents and mental health professionals to develop an Internet-based cognitive-behaviour therapy program for adolescents with anxiety disorders*

Thank you for your comments. We have added to the methods section the additional details around why parental consent was not required. We have also highlighted the section on page 7 which identified which institutional Research Ethics Boards approved the study protocol including recruitment and informed consent procedures.

Sincerely,

Lori Wozney and Amanda S Newton